Author's response to reviews

Title: Prevention of coronary heart disease in people with severe mental illnesses: A qualitative study of patient and professionals' preferences for care

Authors:

Christine A Wright (c.wright@medsch.ucl.ac.uk)
David PJ Osborn (d.osborn@medsch.ucl.ac.uk)
Irwin Nazareth (i.nazareth@pcps.ucl.ac.uk)
Michael B King (m.king@medsch.ucl.ac.uk)

Version: 3 Date: 20 April 2006

Author's response to reviews:

Dear Sir

I attach a revised version of the manuscript which has now been accepted for publication in BMC Psychiatry. This version of the manuscript incorporates the formatting changes requested in your email dated 12th April - i.e. the title 'Box 1' has been amended to 'Table 1', with similar adjustments to all subsequent boxes (4 in total). I have also ensured that all references to the boxes/tables in the text have been amended to reflect the revised titles.

Kind regards
Dr Christine Wright